Author’s response to reviews

Title: Seroprevalence for dengue virus in a hyperendemic area and associated socioeconomic and demographic factors using a cross-sectional design and a geostatistical approach, state of São Paulo, Brazil.

Authors:

Francisco Chiaravalloti-Neto (franciscochiara@usp.br)
Rafael Alves da Silva (rafa_m112@hotmail.com)
Nathalia Zini (nathalia_zini@hotmail.com)
Gislaine Celestino Dutra da Silva (gislaine cds@gmail.com)
Natal Santos da Silva (natalss@gmail.com)
Maisa Carla Pereira Parra (maisapparra@hotmail.com)
Margarteh Regina Dibo (dibomargareth8@gmail.com)
Cassia Fernanda Estofolete (cassiafestofolete@gmail.com)
Eliane Aparecida Fávaro (eliane_favaro@yahoo.com.br)
Karina Rocha Dutra (karina_rocha10@hotmail.com)
Manlio Tasso Oliveira Mota (manliotasso@gmail.com)
Georgia Freitas Guimarães (gfggeorgina@gmail.com)
Ana Carolina Bernades Terzian (anacarolinaterzian@gmail.com)
Marta Blangiardo (m.blangiardo@imperial.ac.uk)
Mauricio Lacerda Nogueira (mnogueira@famerp.br)

Version: 2 Date: 05 May 2019

Author’s response to reviews:

Answers to Editor

We presented below the questions raised by the Editor and our answers.

Questions
1. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Answer: We included the phrase “The funding agencies did not play any role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.” In the Funding section of the manuscript.

2. Please state in the cover letter whether the map depicted in figure 1 is your own or taken from another source.

Answer: We included the phrase “The map depicted in Figure 1 was elaborated by ourselves.” In the end of the 5th paragraph of the Cover letter.

3. Please remove any additional files you do not want published alongside your manuscript, including cover/response letters.

Answer: We did!

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: We did!

05/05/2019

Francisco Chiaravalloti Neto

Correspondent author